**Author's response to reviews**

**Title:** Total Exposure Study Analysis Consortium: a cross-sectional study of tobacco exposures

**Authors:**

Andrew W Bergen (andrew.bergen@sri.com)
Ruth Krasnow (ruth.krasnow@sri.com)
Harold S Javitz (harold.javitz@sri.com)
Gary E Swan (gswan@stanford.edu)
Ming D Li (ml2km@virginia.edu)
James W Baurley (baurley@biorealmresearch.com)
Xiangning Chen (xchen@vcu.edu)
Lenn Murrelle (lenn.murrelle@venebio.com)
Barbara Zedler (barb.zedler@venebio.com)

**Version:** 2  **Date:** 8 September 2014

**Author's response to reviews:**

I have added line numbering, made more prominent the name of the IRB in the first sentence of Methods, and added my postal contacts.